Author’s response to reviews

Title: Prenatal exposure to perfluoroalkyl and polyfluoroalkyl substances and childhood atopic dermatitis: a prospective birth cohort study

Authors:

Qian Chen (chenqian987@gmail.com)
Rong Huang (qingtian_1119@163.com)
Li Hua (18818237316@163.com)
Yifeng Guo (guoyifeng2014@qq.com)
Lisu Huang (huanglisu@126.com)
Yanjun Zhao (13524813362@163.com)
Xia Wang (wxsjtuge@gmail.com)
Jun Zhang (junjimzhang@sina.com)

Version: 3 Date: 23 Dec 2017

Author’s response to reviews:

Dear Editor,

Thank you for your comments on our manuscript titled “Prenatal exposure to perfluoroalkyl and polyfluoroalkyl substances and childhood atopic dermatitis: a prospective birth cohort study” (ENHE-D-17-00244). These comments helped us improve our manuscript, and provided important guidance for future research.

We have addressed the editor’s suggestions to the best of our abilities, and revised the manuscript accordingly. We hope this meets the requirements for a publication.
Reviewer reports:

Editor's comments: The authors did not address the reviewer comments fully. We attach a copy of the MS with our comments.

Response: We have revised the manuscript as suggested. Revisions were highlighted in red.

Should further revision be deemed necessary, please let me know. We really appreciate your time and efforts.

Yours sincerely,

Jun Zhang